Author's response to reviews

Title: The prevalence of hepatitis B virus markers in a cohort of students in Bangui, Central African Republic

Authors:

Narcisse P Komas (npkomas@yahoo.fr)
Souleyman B Sépou (sepou_junior@yahoo.fr)
Alexandre Manirakiza (amanirak@yahoo.fr)
Josiane Léal (carmelle_leal@yahoo.fr)
Aubin Béré (berostau@yahoo.fr)
Alain Le Faou (a.lefaou@chu-nancy.fr)

Version: 4 Date: 27 July 2010

Author's response to reviews:

To The BMC Series Editorial Production Team,

Formatting changes has been made according to yours suggestions:

1 - Major revisions

# Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system

Response: As the multivariate analysis do not contribute to the interpretation of data, we think it is not necessary to include the corresponding table in the text, and thus that it may be omitted.

2 - Minor revisions

# Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Response: We removed all highlighting and tracking from the manuscript

# Abbreviations: Please format these in to a paragraph.

Response: We formatted abbreviations into a paragraph

# Authors' contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points.

Response: We also formatted Authors’ contributions into a paragraph and suppressed all bullet points

# Word count: please delete these.

Response: We deleted word count at the end of the manuscript
# Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: We checked typography for any typographical errors